Author's response to reviews

Title: Expression analysis of Mammaglobin A (SCGB2A2) and Lipophillin B (SCGB1D2) in more than 300 human tumors and matching normal tissues reveals their co-expression in gynecologic malignancies

Authors:

Menelaos Zafrakas (mzafrakas@gmail.com)
Beate Petschke (b.petschke@gmx.de)
Andreas Donner (adonner@ukaachen.de)
Florian Fritzsche (florian.fritzsche@charite.de)
Glen Kristiansen (glen.kristiansen@charite.de)
Ruth Knuechel (rknuechel-clarke@ukaachen.de)
Edgar Dahl (edahl@ukaachen.de)

Version: 4 Date: 20 March 2006

Author's response to reviews:

There were no further objections or suggestions from the two reviewers. We followed the instructions made by the BioMed Central Editorial Team.